

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Tumen, Semih; Zeydanli, Tugba

Working Paper
Social Interactions in Job Satisfaction

IZA Discussion Papers, No. 9505

Provided in Cooperation with:

IZA - Institute of Labor Economics

Suggested Citation: Tumen, Semih; Zeydanli, Tugba (2015): Social Interactions in Job Satisfaction, IZA Discussion Papers, No. 9505, Institute for the Study of Labor (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/125020

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





IZA DP No. 9505

Social Interactions in Job Satisfaction

Semih Tumen Tugba Zeydanli

November 2015

Forschungsinstitut zur Zukunft der Arbeit Institute for the Study of Labor

Social Interactions in Job Satisfaction

Semih Tumen

Central Bank of the Republic of Turkey and IZA

Tugba Zeydanli

Collegio Carlo Alberto

Discussion Paper No. 9505 November 2015

ΙΖΑ

P.O. Box 7240 53072 Bonn Germany

Phone: +49-228-3894-0 Fax: +49-228-3894-180 E-mail: iza@iza.org

Any opinions expressed here are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but the institute itself takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The Institute for the Study of Labor (IZA) in Bonn is a local and virtual international research center and a place of communication between science, politics and business. IZA is an independent nonprofit organization supported by Deutsche Post Foundation. The center is associated with the University of Bonn and offers a stimulating research environment through its international network, workshops and conferences, data service, project support, research visits and doctoral program. IZA engages in (i) original and internationally competitive research in all fields of labor economics, (ii) development of policy concepts, and (iii) dissemination of research results and concepts to the interested public.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ABSTRACT

Social Interactions in Job Satisfaction*

The literature documents that job satisfaction is positively correlated with worker performance and productivity. We examine whether aggregate job satisfaction in a certain labor market environment can have an impact on individual-level job satisfaction. If the answer is yes, then policies targeted to increase job satisfaction can increase productivity not only directly, but through spillover externalities too. We seek an answer to this question using two different data sets from the United Kingdom characterizing two different labor market environments: Workplace Employment Relations Survey (WERS) at the workplace level (i.e., narrowly defined worker groups) and British Household Panel Survey (BHPS) at the local labor market level (i.e., larger worker groups defined in industry x region cells). Implementing an original empirical strategy to identify spillover effects, we find that one standard deviation increase in aggregate job satisfaction leads to a 0.42 standard deviation increase in individual-level job satisfaction at the workplace level and 0.15 standard deviation increase in individual-level job satisfaction at the local labor market level. These social interactions effects are sizable and should not be ignored in assessing the effectiveness of the policies designed to improve job satisfaction.

JEL Classification: C31, D62, J28

Keywords: job satisfaction, social interactions, spillovers, hierarchical model, WERS, BHPS

Corresponding author:

Semih Tumen Research and Monetary Policy Department Central Bank of the Republic of Turkey Istiklal Cad. No:10 06100 Ulus, Ankara Turkey

E-mail: semih.tumen@tcmb.gov.tr

_

We thank Olivier Bargain, Arnaud Chevalier, Andrew Clark, Paul Devereux, Steve Durlauf, Hakan Ercan, Roxana Gutiérrez- Romero, Nicolas Jacquemet, Murat Kirdar, Peter Kuhn, Alex Michaelides, Natalia Montinari, Claudia Senik, Pedro Vicente, seminar participants at the Central Bank of the Republic of Turkey, Galatasaray University, Paris School of Economics, Universidade Nova de Lisboa, TOBB-ETU, the participants of the EDE-EM Summer Meeting in Paris, Asian Meeting of the Econometric Society in Delhi, Turkish Economic Association Annual Conference in Izmir, Workshop on Social Economy in Bologna, Royal Economic Society Conference in Manchester, Applied Economics Meeting in Gran Canaria, and European Association of Labor Economists Conference in Ljubljana for useful suggestions. We are particularly grateful to Martin Kahanec (the Editor) and two anonymous referees for very helpful comments. Tugba Zeydanli gratefully acknowledges financial support from the European Doctorate in Economics - Erasmus Mundus. The views expressed here are of our own and do not necessarily reflect those of the Central Bank of the Republic of Turkey. All errors are ours.

1 Introduction

Job satisfaction is a direct measure of utility an employed worker derives from his/her current job [Clark and Oswald (1996)]. It has extensive behavioral consequences. For example, job satisfaction is a significant determinant of labor market mobility—in particular, the quitting behavior.¹ It is also shown to be related to relative pay comparisons among peer groups in the workplace.² Most importantly, and this is mainly why labor economists should be interested in job satisfaction, it is documented to have a positive correlation with labor productivity and worker performance.³ In particular, Boeckerman and Ilmakunnas (2012) document that job satisfaction has a causal effect on productivity.⁴ To get the feel of the magnitude, Boeckerman and Ilmakunnas (2012) find that one standard deviation increase in job satisfaction within the plant increases productivity per hours worked by 6.6 percent.

Although several aspects of job satisfaction have been studied extensively in the empirical literature, whether there exist spillover externalities in job satisfaction—i.e., whether individual-level job satisfaction is affected by the aggregate job satisfaction in a certain labor market environment—or not remains as an unanswered question. This is a relevant question because job satisfaction is often associated with workplace attitudes such as involvement in the organization, relatedness with co-workers/customers/managers, attachment, motivation, shirking, tendency to slow down work, absenteeism, etc. These attitudes form a workplace "atmosphere" and jointly contribute to the formation of worker satisfaction and performance. Therefore, the aggregate job satisfaction level in a certain work environment can be regarded as a "social" variable and may, in turn, affect individual-level job satisfaction.

Our ultimate goal in this paper is to investigate if there exist any visible footprints of social interactions in job satisfaction. Answering this question is also important for policy. If there

¹See, e.g., Freeman (1978), Akerlof et al. (1988), Clark et al. (1998), and Clark (2001).

²See, for example, Clark et al. (2009) and Card et al. (2012).

³Other studies documenting this positive relationship include, but are not limited to, Iaffaldano and Muchinsky (1985), Ostroff (1992), Brown and Peterson (1994), Ryan et al. (1996), Sloane and Williams (2000), Argyle (2001), Judge et al. (2001), Harter et al. (2002), Schneider et al. (2003), Patterson et al. (2004), Green and Tsitsianis (2005), Otis and Pelletier (2005), Christen et al. (2006), Ghinetti (2007), and Wegge et al. (2007). Zelenski et al. (2008) and Oswald et al. (2013) argue that happiness and life satisfaction are also positively correlated with productivity.

⁴The direction of the causal relationship between productivity and job satisfaction has been controversial in the literature. However, recent evidence suggests that the direction of the relationship goes from job satisfaction to labor productivity.

exist positive spillovers in job satisfaction, then policies targeted to increase job satisfaction can boost productivity not only directly, but through spillover externalities too. When these social interactions effects are sizable, ignoring them may lead to mis-assessment of the effectiveness of the policies designed to improve job satisfaction in various work environments.

We perform our empirical analysis at two aggregation levels using two different data sets from the United Kingdom. First, we use the Workplace Employment Relations Survey (WERS) to test the existence of job satisfaction spillovers at the workplace level (or establishment-level).⁵ In the workplace-level analysis, the reference group that the social forces are effective is the set of workers in each workplace. Second, we use the British Household Panel Survey (BHPS) to form industry × region cells for the purpose of testing the existence of spillovers at the local labor market level. In this second exercise, we try to capture more general social effects in larger reference groups. The main purpose is to focus on social processes that involve collective aspects of community and work life. In both of these exercises, we concentrate on estimating the correlation between the group-level and individual-level job satisfaction scores, controlling for a large set of observed covariates. Drawing a distinction between the workplace and local labor market level analyses is useful, because it will allow us to make precise statements on whether it is more effective to enforce job satisfaction policies at the establishment level (i.e., as firm-specific policies) or local labor market level (i.e., in the form of broader institutional measures).

Our econometric framework will be a version of the canonical linear-in-means model, which is a base for the bulk of empirical work on social interactions.⁶ The main problem with the linear-in-means model is that it necessitates employing a carefully-designed identification strategy to separate endogenous effects from the contextual effects [Manski (1993)]. It will perhaps be useful at this point to clearly define the terms "endogenous social effects" and "contextual social effects." The endogenous effect refers to the effect of the group-level outcome on the

⁵The terms "workplace" and "establishment" will be used interchangeably throughout the paper.

⁶See Blume et al. (2011) for an in-depth background information on linear-in-means models, including a comprehensive discussion on micro-foundations and econometric identification. Also see Blume and Durlauf (2001), Brock and Durlauf (2001b), and Soetevent (2006).

⁷See also Manski (2000) and Brock and Durlauf (2001b) for a more detailed discussion of the different types of social interactions effects.

individual-level outcome. Within the context of our paper, this corresponds to the effect of the group-level mean of job satisfaction on the individual-level job satisfaction. The contextual effect, on the other hand, refers to the effect of the group-level counterparts of the individual-level observables on the individual-level outcome variable; in our paper, this corresponds to the effect of, say, group-level average age or average education on the individual-level job satisfaction score.

At the center of our identification strategy lies an insight from the hierarchical (or multilevel) statistical models of social processes: social groups describe "ecologies" in which decisions are made and matter because different ecologies induce different mappings from the individual determinants of these decisions to the associated outcomes [Raudenbush and Sampson (1999)]. Based on this conceptualization, we construct an empirical model in which contextual effects (i.e., the "ecologies" in our model) alter the coefficients linking individual characteristics to outcomes. This corresponds to allowing for multiplicative interactions between the contextual effects and the remaining explanatory terms within the linear-in-means model. We formally show that introducing these cross-product terms induces non-linearities that resolve the reflection problem Manski (1993) describes [see Section 3]. Such a setup secures the econometric identification of social interactions and enables us to separate endogenous effects from the contextual effects [Blume and Durlauf (2005)]. Although, this approach is rather simple and intuitive, it is surprisingly under-utilized in the literature.

There are two more potential threats to identification. The first one is the possibility of sorting into reference groups based on unobserved factors [Manski (1993)]. More specifically, if there exist group-level unobserved factors that determine individual-level job satisfaction and are also correlated with the group-level job satisfaction, then the resulting estimates would be biased. Our empirical approach also allows us to address this problem, at least partially, by controlling for group-level unobservables in various ways. And, second, it is well-documented in the literature that the relative income structure within the reference group is an important determinant of the job satisfaction level in the peer group [Card et al. (2012)]. We also control for the pay-comparison effects in our calculations.

We find that one standard deviation increase in aggregate job satisfaction level leads to a 0.42 standard deviation increase in individual-level job satisfaction score at the workplace level and a 0.15 standard deviation increase in individual-level job satisfaction score at the local labor market level. In other words, we report that statistically significant job satisfaction spillovers exist both at the establishment level and local labor market level; and, the estimated spillovers are approximately three times larger at the establishment level than those at the local labor market level. These estimates can be restated in terms of the social multiplier: the corresponding social multipliers are $[1/(1-0.42)\approx]$ 1.72 and $[1/(1-0.15)\approx]$ 1.18 at the workplace and local labor market levels, respectively. Simple calculations yield the result that the Boeckerman and Ilmakunnas (2012) estimates—which say that one standard deviation increase in job satisfaction within the plant increases productivity per hours worked by 6.6 percent—would be revised up to 11.4 percent at the workplace level and 7.8 percent at the local labor market level after accounting for the job satisfaction spillovers. To summarize, these results suggest that (1) failing to account for the spillover externalities in job satisfaction may lead us to mis-assess the effectiveness of job satisfaction policies; thus, the policy maker should internalize these externalities, and (2) job satisfaction spillovers are much stronger at the workplace level than local labor market level; therefore, designing/enforcing job satisfaction policies at the workplace level will likely be more effective than implementing such policies at the local labor market level.

We also report estimates for contextual social effects. At the workplace level, we find that individual-level job satisfaction goes up with the fraction of male and older workers in the workplace. At the local labor market level, however, gender and age do not have any statistically significant contextual effect; instead, we only find that individual-level job satisfaction score goes down as the fraction of workers with greater access to promotion opportunities goes up in each industry \times region cell. We also document that there are significant "income comparison effects" at the workplace, but not at the local labor market. In particular, we find that individual-level job satisfaction goes down with (i) average earnings and (ii) fraction of high earners—i.e., those who earn above the median wage within the relevant worker

⁸See Glaeser et al. (2003) for an excellent discussion of the social multiplier concept.

population—in the workplace. We discuss these results further in Section 5, where we also present detailed robustness exercises to prove that our estimates are not excessively sensitive to relaxing some of the main assumptions behind the empirical model.

The plan of the paper is as follows. Section 2 relates and compares our work to the relevant papers in the literature. Section 3 explains the details of the econometric model and the identification strategy we employ. Section 4 provides an overview of the data sets we use and justifies the construction of our reference groups in different work settings. Section 5 presents the estimates, discusses in detail the results, performs robustness checks, and elaborates on the policy implications. Section 6 concludes.

2 Related Literature

Our paper can be placed into the literature investigating social interactions in labor markets. There is a large body of literature testing the existence of peer effects in various labor market outcomes including productivity, wages, absenteeism, and learning (or knowledge spillovers). The results are mixed. For example, using grocery scanner data from a large supermarket chain, Mas and Moretti (2009) perform a field experiment among low-wage earners to analyze if the individual-level effort is influenced by a permanent increase in the productivity of co-workers and find reasonably large peer effects. Falk and Ichino (2006) study the behavior of high school students performing a simple task in a laboratory experiment to understand if individual-level performances are directly affected by the existence of other individuals doing the same task and they also document moderate peer effects. Ichino and Maggi (2000) find that group-level peer absenteeism increases individual absenteeism. In a field study, Bandiera et al. (2009) find that individual-level productivity responds to the skill-level of a friend working nearby, but does not respond to the skill-level of a non-friend working around. Guryan et al. (2009), on the other hand, find employing a random assignment exercise on a golf tournament data that individual-level performance is not influenced by the playing partners' ability. Cornelissen et al. (2013) report only small peer effects in wages among co-workers.⁹ While

⁹Here, "peer effects in wages" refer to the idea that peer-induced productivity increases could be rewarded in the form of higher wages.

Azoulay et al. (2010) and Jackson and Bruegmann (2009) document significant knowledge spillovers among co-workers, Waldinger (2012) shows that those spillovers are weak, if they ever exist.

There are also several papers investigating contagion effects in subjective well-being measures. Using Chinese rural survey data, Knight and Gunatilaka (2009) examine whether happiness is infectious or not at the village level. Their results show that happiness is infectious in narrowly-defined reference groups. They exploit the panel feature of their data set to remove the reflection problem and identify the relevant social effects. Papers in the psychology literature also find that happiness might be contagious in small environments. However, these studies do not address the reflection problem, which might bias the results. Tumen and Zeydanli (2014), on the other hand, find that these contagion effects might disappear in more broadly defined reference groups.

Our paper differs from this body of work and contributes to the related literature in three ways. First, this is the first paper in the literature estimating spillover effects in job satisfaction. We show that there exist statistically and economically significant job satisfaction spillovers in various work environments. Second, we show that the degree of these spillover externalities may change at different aggregation levels. Using two different data sets from the United Kingdom, we construct our reference groups at two aggregation levels: workplace level and local labor market level. The former defines peer effects in narrowly defined work settings, while the latter defines the social environment in larger ecological settings that embed more general aspects of community and working life. We document that the job satisfaction spillovers exist in both environments; but, they are much stronger at the workplace level than local labor market level. We further argue that this may have important policy implications. And, third, motivated by the hierarchical models of social processes, we develop an original identification strategy to separate endogenous effects from the contextual effects, controlling for group-level unobserved heterogeneity.

¹⁰See, e.g., Hatfield et al. (1994), Sato and Yoshikawa (2007), and Fowler and Christakis (2008).

There are also several papers criticizing the empirical literature on peer effects. ¹¹ In particular, Angrist (2014) has pointed out that many papers in the literature falsely interpret the observed correlations between individual- and group-level outcomes as causal relationships. He argues that the instrumental variable (IV) estimates reported in this literature "often ... produce findings that look like a peer effect, even in a world where behavioral influences between peers are absent." In this paper, we do not use the IV approach; instead, we rely on a nonlinear model of social interactions—which is motivated by hierarchical statistical models—to obtain econometric identification. The non-linear models are not free of problems, either. The most common criticism is that, most of the time, the non-linear structure used to identify peer effects is hard to justify. The non-linear specification that we use in this paper has two appealing features. First, the non-linearity is simply obtained by including certain interaction terms between the regressors of a standard linear-in-means model. The use of interaction terms is quite common in regression analysis and are never regarded as strange or unjustified. Second, the inclusion of interaction terms into linear-in-means models is theoretically justified by the "social ecologies" viewpoint in a strand of the sociology literature. ¹² See Raudenbush and Sampson (1999) and Blume and Durlauf (2005) for further motivation and references. Next, we present the details of our non-linear model of social interactions and describe our identification strategy.

3 Model and Theoretical Background

The econometric framework that we employ in this paper is a version of the canonical linear-in-means model of social interactions. Our ultimate goal is to estimate social interactions in job satisfaction. In particular, we would like to estimate the effects of (1) group-level job satisfaction—the "endogenous social effect"—and (2) group-level exogenous characteristics—the "contextual effects"—on individual-level job satisfaction, controlling for group-level heterogeneity. The linear-in-means model of social interactions is plagued with the well-known "reflection problem," which masks the econometric identification of social interactions [Man-

¹¹See, e.g., Deaton (1990), Moffitt (2001), and Angrist (2014).

¹²The origins of the term "social ecologies" or "ecological settings" goes back to the Chicago School of Sociology in the 1920s [see, e.g., Cavan (1983)].

ski (1993)]. The simplest way to resolve this issue is to use an appropriately formulated instrumental variables strategy. When an instrument is not available, it is necessary to invoke non-linearities to identify social interactions [Brock and Durlauf (2001a), Blume et al. (2011)].

In this paper, we use an empirical strategy that allows us to convert the standard linear model into a nonlinear one. The motivation comes from the hierarchical models of social processes. This hierarchical structure secures identification of social interactions via introducing cross-product terms into the standard model. This section provides a detailed description of our econometric model for the purpose of familiarizing the reader with the basic concepts we frequently mention throughout the paper. Section 3.1 presents our empirical model and the associated technical issues (i.e., the reflection problem) including a formal statement of the conditions required to identify social interactions. Section 3.2 describes our hierarchical model and assesses in detail how we achieve identification.

3.1 The Empirical Model of Social Interactions

Each individual $i \in \mathcal{I}$ is a member of a group $g \in \mathcal{G}$, where \mathcal{I} is the number of individuals in the worker population and \mathcal{G} is the number of groups, with $\mathcal{I} > \mathcal{G}$. The following linear-in-means equation is an empirical tool commonly used in the literature:

$$\omega_{i_g} = \beta_0 + \beta_1 \mathbf{X}_{i_g} + \beta_2 \mathbf{Y}_g + Jm_g + u_g + \epsilon_{i_g}, \tag{3.1}$$

where the dependent variable, ω_{i_g} , is the individual-level job satisfaction for person i in group g, X_{i_g} is a vector of individual-level observed characteristics of i in group g, Y_g is a vector of group-level observed characteristics of group g, $m_g = \mathbb{E}[\omega_{i_g}|g, F_{i_g}]$ is the mean job satisfaction in group g, u_g is a group-level unobserved factor common across the members of group g, and ϵ_{i_g} is a random error term with $\mathbb{E}[\epsilon_{i_g}|g, F_{i_g}] = 0$. In our notation, F_{i_g} corresponds to the empirical distribution of individuals in group g and this distribution is possibly different for each group. The distinction between β_2 (contextual effects) and J (endogenous effect) is the key notion in this model. The former measures the effect of exogenous group-level variables on the individual-level outcome, while the latter measures the effect of endogenous group-level

outcome on the individual-level outcome. Our ultimate goal is to clearly distinguish β_2 from J and to separately identify the effects of group-level variables on the individual-level outcome. However, econometric identification is a problematic issue in this standard setting. In what follows, we shut down the group-level unobserved effect u_g for notational simplicity. It will reappear in our final equation.

To define the identification problem, we take the conditional mathematical expectations of both sides of the linear-in-means equation, where the conditioning is on g and F_{ig} , for all i and g. This gives us

$$m_g = \beta_0 + \beta_1 \mathbf{X}_g + \beta_2 \mathbf{Y}_g + J m_g, \tag{3.2}$$

where $X_g = \mathbb{E}[X_{i_g}|g, F_{i_g}]$. X_g can be named as the group-level mean of individual-level observed characteristics and it may or may not coincide with Y_g . Notice that m_g appears in both sides of Equation (3.2). Solving for m_g yields the result that

$$m_g = \frac{\beta_0}{1 - J} + \frac{\beta_1}{1 - J} \mathbf{X}_g + \frac{\beta_2}{1 - J} \mathbf{Y}_g.$$
 (3.3)

The reflection problem states that if the dimensions of the vectors X_g and Y_g are the same, then linearity masks the econometric identification of the (endogenous) social interactions parameter J.

To formalize this statement, we plug Equation (3.3) into Equation (3.1), which gives us the estimating equation

$$\omega_{i_g} = \frac{\beta_0}{1 - J} + \beta_1 \boldsymbol{X}_{i_g} + \frac{J\beta_1}{1 - J} \boldsymbol{X}_g + \frac{\beta_2}{1 - J} \boldsymbol{Y}_g + \epsilon_{i_g}. \tag{3.4}$$

When the reflection problem is in effect, J and β_2 cannot be distinguished from each other, which implies that social interactions cannot be identified. To see this, set $\mathbf{X}_g = \mathbf{Y}_g$, which

yields the equation

$$\omega_{i_g} = \frac{\beta_0}{1 - J} + \beta_1 \boldsymbol{X}_{i_g} + \frac{J\beta_1 + \beta_2}{1 - J} \boldsymbol{Y}_g + \epsilon_{i_g}. \tag{3.5}$$

It is obvious that, in this equation, it is impossible to separate J from β_2 econometrically. One solution is the existence of an additional X_g which is not in Y_g . If such an X_g exists, then endogenous social interactions (J)—and also all the other model parameters—are identified by applying simple ordinary least-squares method on Equation (3.4). In other words, one individual-level variable, the mean of which cannot be regarded as a group-level variable, is required for identification of social interactions.

Unfortunately, most of the large data sets—such as BHPS, GSOEP, WERS, etc.—do not include a variable X_g that can naturally fit into the IV definition provided above. One popular alternative to IV is to introduce non-linearities into the linear-in-means specification. To demonstrate how non-linearities secure identification, we modify the standard model as follows:

$$\omega_{i_q} = \beta_0 + \beta_1 \mathbf{X}_{i_q} + \beta_2 \mathbf{Y}_g + J\phi(m_g) + \epsilon_{i_q}, \tag{3.6}$$

where $\phi(\cdot)$ has non-zero second derivatives; that is, it is a legitimate nonlinear function. Without loss of generality, we assume also that $\phi(\cdot)$ is invertible. Again, taking the conditional mathematical expectations of both sides and rearranging the terms in such a way that the terms with m_q appears on the left and the rest of the variables on the right, we get the equation

$$\Phi(m_g) = \beta_0 + \beta_1 \mathbf{X}_g + \beta_2 \mathbf{Y}_g, \tag{3.7}$$

where $\Phi(m_g) = m_g - J\phi(m_g)$. The functions $\phi(\cdot)$ and $\Phi(\cdot)$ has the same properties, therefore we can invert $\Phi(\cdot)$ to get

$$m_q = \Phi^{-1} \left(\beta_0 + \beta_1 \boldsymbol{X}_q + \beta_2 \boldsymbol{Y}_q \right) \tag{3.8}$$

and plugging this into the original estimating equation we get

$$\omega_{i_q} = \beta_0 + \beta_1 \boldsymbol{X}_{i_q} + \beta_2 \boldsymbol{Y}_g + J\phi \left[\Phi^{-1} \left(\beta_0 + \beta_1 \boldsymbol{X}_g + \beta_2 \boldsymbol{Y}_g \right) \right] + \epsilon_{i_q}. \tag{3.9}$$

In such a setting, we can identify β_2 and J separately without a further need for an exclusion restriction (or an IV). One problem with this framework is that there is no systematic way of choosing the functional form of $\phi(\cdot)$. In this paper, we propose an estimation strategy that introduces a systematic way to embed non-linearities into the standard empirical specification. To be specific, we construct a hierarchical model, which has the additional advantage of being consistent with our definition and conceptualization of social interactions in job satisfaction.

3.2 The Hierarchical Model

Suppose that the Equation (3.1) is modified as follows:

$$\omega_{i_g} = \alpha_0 \left(\mathbf{Y}_g \right) + \alpha_1 \left(\mathbf{Y}_g \right) \mathbf{X}_{i_g} + \alpha_J \left(\mathbf{Y}_g \right) m_g + u_g + \epsilon_{i_g}. \tag{3.10}$$

In words, the coefficients α_0 , α_1 , and α_J are stated as functions of the contextual variables, \mathbf{Y}_g , which define the "social context." In other words, the contextual variables describe the properties of the environments that the individuals live in. Setting up the regression coefficients in this way implies that social groups describe ecologies in which decisions are made and matter because different ecologies induce different mappings from the individual determinants of these behaviors and choices.¹³ To convert this setting into an empirical equation that we can estimate, we make the following simplifying assumptions:

$$\alpha_0 \left(\mathbf{Y}_g \right) = \beta_0 + \beta_2 \mathbf{Y}_g,$$

$$\alpha_1\left(\boldsymbol{Y}_g\right) = \beta_1 + \boldsymbol{b}\boldsymbol{Y}_{\boldsymbol{g}},$$

$$\alpha_J(\mathbf{Y}_g) = J + \pi \mathbf{Y}_g.$$

¹³As an example, suppose that we have two contextual variables in a workplace environment: age and gender. In such a case, whether the environment is dominated by young versus old workers or female versus male workers or not really matters. In other words, the determinants of job satisfaction will depend on the "atmosphere" formed by these team or group attitudes.

Plugging these expressions into Equation (3.10) yields

$$\omega_{i_g} = \beta_0 + \beta_1 \mathbf{X}_{i_g} + \beta_2 \mathbf{Y}_g + J m_g + \pi \mathbf{Y}_g m_g + \mathbf{Y}_g' \mathbf{B} \mathbf{X}_{i_g} + u_g + \epsilon_{i_g}, \tag{3.11}$$

where B is the matrix form of the coefficient vector b. This equation looks very similar to our original linear-in-means specification except that we include interaction terms in the form of cross products motivated by the hierarchical model.

Note that the nature of the unobserved group-level effect u_g is a crucial issue. There are two alternatives: random effects versus fixed effects. If the true unobserved group-level effect can be controlled for up to a random error term, then a common way to resolve this issue is to assume that u_g is itself random rather than fixed. Assuming that u_g is random is equivalent to saying that it is uncorrelated with the regressors. However, group-level unobserved factors can easily be argued to be correlated with, say, group-level job satisfaction. When this is the case, not being able to control group fixed effects will yield erroneous results. In our baseline analysis, we assume that u_g is a random term. We then relax this assumption and control for group-level fixed effects to check if the results differ [see Section 5]. We also cluster standard errors at the group level, which means that we account for within-group correlations in the error structure.

To demonstrate how this model is identified, we take the conditional mathematical expectations of both sides, as before, and solve the resulting equation for m_g , which gives us

$$m_g = \frac{\beta_0 + \beta_1 \mathbf{X}_g + \beta_2 \mathbf{Y}_g + \mathbf{Y}_g' \mathbf{B} \mathbf{X}_g}{1 - J - \pi \mathbf{Y}_g}.$$
 (3.12)

Notice that, very similar to the motivation behind the nonlinear model, this model also introduces non-linearity between m_g and the other regressors, even when $X_g = Y_g$. There is no need for an exclusion restriction and econometric identification of social influences is immediate, given standard conditions on individual- and group-level observed covariates [see Blume and Durlauf (2005) for further details]. This formulation is consistent with our hypothesis and our definition of social interactions.

At the end, we estimate Equation (3.11) to separately identify β_2 and J. In this setup, the endogenous social effect is $J + \pi \bar{Y}_g$, where \bar{Y}_g are the sample means of group-level variables, i.e., the endogenous effect is no more J since we have cross-product terms in the regressions. The same logic applies to the contextual effects we estimate. The estimates we report and discuss in Section 5 directly refer to these "marginal effects."

4 Data and Reference Groups

In this section, we provide a detailed description of the two data sets we use in our empirical analysis: Workplace Employment Relations Survey and British Household Panel Survey. Both of these surveys are nationally representative data sets for the United Kingdom and provide establishment-level and individual-level labor market information, respectively. We also describe in detail the construction of our reference groups for both of these data sets. We focus on the 2004 editions of both data sets.

4.1 Workplace Employment Relations Survey (WERS)

WERS is a national survey of British employees constructed for the purpose of collecting information on employment relations in Britain.¹⁴ The survey provides information about workers, working conditions, and industrial relations from all sectors except primary industries and private households with domestic staff. WERS 2004—the version that we use in our analysis—is the fifth among a series of surveys. Previous surveys are conducted in 1980, 1984, 1990, and 1998. In the 2004 cross-section, there are around 2,300 workplaces, 1,000 employee representatives, and 22,500 employees.

We construct the job satisfaction scores using the following seven question in the WERS-2004 data set. How satisfied are you with

- 1) the sense of achievement you get from work?
- 2) the scope for using your own initiative?

¹⁴The most recent version of this data set has been co-sponsored by the Department for Business, Innovation and Skills (BIS), Acas, the Economic and Social Research Council (ESRC), the UK Commission for Employment and Skills (UKCES), and the National Institute of Economic and Social Research (NIESR).

- 3) the amount of influence you have over the job?
- 4) the training you receive?
- 5) the amount of pay you receive?
- 6) the job security?
- 7) the work itself?

The responses are based on a five-point scale with 1 representing "very satisfied," 2 "satisfied," 3 "neither satisfied nor dissatisfied," 4 "dissatisfied," and 5 "very dissatisfied." For each of the seven questions listed above, we construct a binary variable for the positive responses—taking the value 1 for the "very satisfied" or "satisfied" responses and 0 otherwise—and, then, we construct a sum of the seven binary variables for each individual to form an index with values from 0 to 7 [see also Jones et al. (2009), Jones and Sloane (2010), and Mumford and Smith (2013)]. We use this aggregate measure as the "job satisfaction score" in our analysis. The average job satisfaction score in our sample is 4.20 and the standard deviation is 2.13. The BHPS data set, which we describe in the following subsection, has a 1–7 scale constructed based on different principles. For the sake of comparability of the estimates, we standardize the main job satisfaction measures in both WERS and BHPS around zero mean and unit variance. Thus, the dependent variable in our analysis will be the "standardized job satisfaction."

We control for a large set of individual- and job-related characteristics. To achieve consistency between the two data sets, we construct the WERS variables similar to their counterparts in the BHPS data set. After excluding missing information on our control variables and dropping workplaces with less than two employees, the WERS data set includes 1,673 workplaces/establishments and, in each workplace, up to 25 randomly-chosen employees taking the questionnaire. We start with describing the education variables. Since this is a workplace-level data set, "No Qualification" category includes only a very small number of observations; thus, we drop the observations in this category and concentrate on the following

¹⁵Although Mumford and Smith (2013) use the six facets of job satisfaction in the WERS, neglecting the training, Jones and Sloane (2010) use all of them. Jones et al. (2009) argue that training is also an important component for job satisfaction. We also include training.

¹⁶For consistency, the individuals in the BHPS data set who work in firms with less than two employees are correspondingly dropped.

education levels: "Higher Degree" (refers to postgraduate education), "First Degree" (refers to college education), "A-level," "O-level" (both referring to different classes of high-school education), and "Vocational Qualification." ¹⁷ Earnings variable in the WERS is reported in 14 pre-specified intervals, ¹⁸ and, following Mumford and Smith (2009), we use the midpoints of these intervals as our earnings variable for each individual. The last interval is open ended, so it does not have a midpoint; instead, we use the mean earnings for the last interval. In our sample, the average weekly log earnings is around 5.7. We also include relative earnings as a dummy variable taking 1 if the employee earns more than the mean earnings in the sample. We categorize the job status under three sector categories: private sector job, public sector job, and other. An establishment size variable is generated from the question of "Currently, how many employees do you have on the payroll at this establishment?" The answer varies from 5 to 10,000. We construct three variables for establishment size; small establishment (less than 50 employees), medium-size establishment (between 50 and 200 employees), and large establishment (more than 200 employees). Working hours are simply represented as a dummy variable taking 1 if the actual hours worked is above the sample mean and 0 otherwise.

Among 20,035 employees and 1,673 establishments in our sample, the average age is 42, 47 percent are male, and 68 percent are married. Higher degree has the lowest fraction, whereas vocational qualifications have the highest. 46 percent of the employees are union members. 55 percent of the workplaces are publicly owned. Regarding the establishment size, the shares of small, medium, and large establishments are 0.32, 0.32, and 0.36, respectively. See Table (1) for detailed summary statistics for our WERS sample.

4.2 British Household Panel Survey (BHPS)

The BHPS provides information on individual-, household-, and job/employer-related characteristics from 1991 to 2008 in England, Scotland, Wales, and Northern Ireland. It yearly follows the same representative sample of households interviewing every adult member of sam-

¹⁷To be concrete, after constructing our sample, we observed that only two individuals remained in the data with no school degrees. We think that dropping these two observations would be more reasonable than generating a new education category called "no degree." Such a category, on the other hand, exists in our BHPS sample since the number of individuals with no degree is non-negligible in the BHPS data.

¹⁸The question of the earnings variable is the following: "How much do you get paid for your job here, before tax and other deductions?"

pled households. Eighteen waves of data are available. To make the two data sets comparable and compatible, we focus on the 2004 cross-section of the BHPS.

The individual-level job satisfaction in the BHPS data set is reported based on a seven-point scale ranging from 1 (not satisfied at all) to 7 (completely satisfied). The employed workers are asked to rate the job satisfaction levels regarding the total income, job security, the actual work itself, and hours worked. The last question about job satisfaction is "Overall, how satisfied or dissatisfied are you with your present job?", which is again measured on the 1–7 scale and used as the "job satisfaction score" in our analysis. As we explain above, we standardize the job satisfaction score around zero mean and unit variance to achieve consistency across the job satisfaction measures we use for the WERS and BHPS data sets.

For the individual-level observed characteristics, we control for gender, age, education level, marital status, earnings, and pay comparisons. We collapse the education-levels into seven broad groups as follows: higher degree refers to postgraduate education, first degree refers to college education, A-level, O-level, and other higher qualification refer to high school graduates of different types (consistent with the education system in the UK), vocational qualification refers to teaching, nursing, commercial, apprenticeship, and the certificate of secondary education (CSE), and, finally, the ones with no qualification. The earnings variable—usual gross pay per month: current job—is recorded as the actual amount received and, thus, we simply take the natural logarithm of this variable in our analysis. We also consider the "taste for working hours" variable. Promotion opportunities is described by the binary variable taking 1 if the worker has access to promotion opportunities and 0 otherwise. The rest of the variables—firm size, job status, relative earnings, and union membership—are constructed similar to their counterparts in our WERS sample.

Table (2) presents the summary statistics of the sample that we use in our analysis. In order to be included into our sample, the respondent have to be employed and report a job satisfaction score. The mean age of the respondents is 40.4. Among the 6,428 observations, 47.4 percent are male, 57.3 percent are married, 4.4 percent have higher degree, 15.8 percent have first

degree, another 12.6 percent have A-level degree, 17.6 percent have O-level degree, 30.8 percent have other higher qualifications, 9.4 percent have vocational qualifications, and the remaining 9.4 percent have no qualifications. Before standardization, the mean job satisfaction score is approximately 5.4 out of 7, with a standard deviation of 1.26. 79 percent are employed in full-time jobs. 63 percent are employed in privately-owned firms. 32.8 percent prefer to work fewer hours. 48.6 percent are employed in small-size firms. See Table (2) for further information on region- and industry-specific details. We generate group-level variables based on our reference groups constructed as industry × region cells. Below we describe how we construct our reference groups both in the WERS and BHPS data sets.

4.3 Reference Groups

Our primary objective is to separately identify endogenous social effects and contextual social effects in job satisfaction within a formal empirical model of social interactions. We conceptualize the social interactions that we estimate as the existence of "spillovers" in the society in the sense that the group-level job satisfaction in one's reference group affects the individual worker's perception of own job satisfaction. We perform this task at two levels with two different data sets from the United Kingdom. First, we use the WERS data set to estimate spillovers at the workplace level. And, second, we use the BHPS data set to estimate job satisfaction spillovers at the local labor market level. The WERS data set captures the social effects among co-workers, who are directly interacting. The BHPS data set, on the other hand, captures social effects among individuals who are potentially interacting indirectly. As Bramoulle et al. (2009) clearly state, this type of social effects is based on the idea that "neighbors in the neighborhood do not affect me directly; what matters is the neighborhood itself."

WERS. The WERS data sets naturally offers establishment-level reference groups; that is, all workers employed in a given establishment constitute the reference group for each of the workers employed in that establishment. There are 1,673 establishments in our WERS sample. Thus, the number of reference groups is 1,673. The average group size is approximately

12 worker per establishment. This setting defines narrow reference groups hypothesizing that social forces operate at the workplace level: workers in a given establishment are exposed to similar work-specific conditions that shape their job satisfaction perceptions. The group-level counterparts of the individual-level variables are constructed taking averages at the workplace level. Similarly, the endogenous social variable (the group-level job satisfaction score) is calculated by averaging the job satisfaction scores within the workplace.

BHPS. For the BHPS data set, we construct industry \times region cells as our reference groups. In terms of our conceptualization of social interactions, this means that we try to capture the social forces that operate among workers who are geographically close to each other and who are potentially exposed to similar local labor market conditions specific to the industries they belong to. This is a common way of constructing reference groups in the empirical social interactions studies, particularly the ones handling large data sets. For example, Luttmer (2005) utilizes the outgoing rotation groups feature of the Current Population Survey and constructs industry \times occupation cells to estimate the neighborhood effects of income on individual-level happiness. Similarly, Ferrer-i-Carbonell (2005) uses the German Socio-Economic Panel and constructs education \times age \times region cells to estimate the impact of the group-level income on individual-level subjective well-being. In a similar context, Glaeser et al. (1996) construct region-specific cells on a lattice to estimate the impact of neighbors' criminal-activity decisions on the agent's own decision to participate in crime. In another example, Stutzer and Lalive (2004) use data from Switzerland cantons and construct canton-level cells to estimate the effect of social norm to work—roughly, the rate of employment in one's neighborhood—on how quickly the unemployed individual finds a job, probably due to social pressure. The examples can be extended further. In all of these papers, large reference groups are constructed to capture the peer influences in broad social settings.

In our BHPS sample, the following twelve regions describe the geographical clustering: 1) London, 2) South East, 3) South West, 4) East Anglia, 5) East Midlands, 6) West Midlands, 7) North West, 8) North East, 9) Yorkshire & Humberside, 10) Wales, 11) Scotland, and 12) Northern Ireland. Nine industry categories are selected at one-digit level as follows: 1) energy

& water supplies, 2) extraction of minerals & manufacture of metal goods, mineral products & chemicals, 3) metal goods, engineering & vehicles, 4) other manufacturing industries, 5) construction, 6) distribution, hotels & catering (repairs), 7) transport & communication, 8) banking, finance, insurance, business services & leasing, and 9) other services. At the end, there are 108 reference groups in our BHPS sample. The average group size is approximately 60 workers per industry × region cell.

5 Results and Discussion

In this section, we present the estimation results, provide a detailed interpretation of the estimates, perform robustness checks, and discuss the policy implications. We use two data sets: WERS and BHPS. In WERS, establishments are the reference groups, whereas, in BHPS, reference groups are defined by the industry × region cells. We group our estimates under three categories: individual-level coefficients, endogenous social effects, and contextual social effects. Individual-level coefficients describe the impact of individual-level observed covariates on the job satisfaction score. The endogenous social effect refers to the effect of the mean job satisfaction—where the mean is calculated at the group level—on the job satisfaction score. The contextual social effect refers to the effect of group-level counterparts of the individual-level covariates on the job satisfaction score. Below we discuss our estimates in detail. Note that both the individual- and group-level job satisfaction scores are standardized around mean zero and unit variance.

It will perhaps be useful to clearly explain how we present the results. We start with a model, which we call the "baseline" case, relying on two strong assumptions: (i) group-level means of the key variables are calculated by including the individual's own outcomes and/or characteristics and (ii) group-level fixed effects are omitted. We acknowledge that these are strong assumptions and can contaminate the estimates. We then proceed by relaxing each of these assumptions. The final results will be the most realistic ones. This line of reasoning will allow us to evaluate the potential biases that would be caused by each of these assumptions.

5.1 Hierarchical Model

We start our analysis by estimating our core model: the hierarchical model of social interactions given by Equation (3.11). The estimations are performed both for the WERS & BHPS data sets. In the baseline specification, we control for group-level effects by defining them as unobserved random effects. We relax this assumption in Section 5.2, where we perform robustness checks. We report the estimates in two forms. First, we document only the marginal effects for WERS and BHPS given in Tables (3) and (4), respectively. The marginal effects are readily interpretable as the social interactions estimates, so they will be at the center of our analysis. Second, we deal with the interaction terms. The coefficients of the interaction terms are useful, because they inform us whether the social interactions effects are heterogeneous or not.

5.1.1 Marginal Effects

Below we separately report the marginal effects for individual-level coefficients, endogenous social effects, and contextual social effects. As usual, the marginal effects evaluate the interaction terms at their respective sample means.

Estimates for the Individual-level Coefficients. Our estimates for the individual-level coefficients are parallel to those reported in the previous empirical literature on the determinants of job satisfaction [see, for example, Clark (1996), Clark and Oswald (1996), and Taylor (2006)]. Specifically, for both WERS and BHPS, we find that females, married workers, younger workers, workers with higher earnings, workers earning more than the median wage earner in the population, workers with greater access to promotion opportunities, and workers employed in smaller establishments are more satisfied jobwise. ¹⁹ The first two columns of Tables (3) and (4) report the estimates of individual-level covariates for the WERS and BHPS data sets, respectively. This paper focuses on estimating spillovers in job satisfaction; thus, the rest of the paper aims to interpret the estimated social effects rather than providing a lengthy discussion of the individual-level covariates.

¹⁹Note that the estimates for the promotion opportunities are only relevant for the BHPS, since the WERS data set does not include a question regarding the promotion prospects of the employees.

Estimates for Endogenous Social Interactions. A group-level variable is endogenous if its individual-level counterpart is the choice variable of interest. Hence, the associated group-level variable can be defined as the effect of other people's behavior on individual-level behavior. This a classic example of spillover externalities. The findings from our benchmark estimates verify that there exist significant positive spillover externalities in job satisfaction; that is, the group-level (i.e., mean) job satisfaction is positively related to individual-level job satisfaction. To put it differently, an individual worker's job satisfaction level tend to be higher in a group of workers who are highly satisfied jobwise. We document these effects for both the WERS and BHPS samples. We find that one standard deviation increase in aggregate job satisfaction level leads to a 0.42 standard deviation increase in individual-level job satisfaction at the workplace level and 0.15 standard deviation increase in individual-level job satisfaction at the local labor market level. The first line of the estimates both in Tables (3) and (4) reports the numbers for endogenous spillovers for the WERS and BHPS data sets, respectively. Job satisfaction is often associated with workplace attitudes such as involvement in the organization, relatedness with co-workers/customers/managers, attachment, motivation, shirking, tendency to slow down work, absenteeism, etc. These attitudes form a workplace "atmosphere" and jointly contribute to the formation of worker satisfaction and performance. Our estimates confirm that the aggregate job satisfaction level in a certain work environment can be regarded as a "social" variable and may, in turn, affect individual-level job satisfaction significantly.

This result suggests that there are huge gains to policy interventions to increase individual-level job satisfaction as there are large positive feedback effects from group-level job satisfaction toward individual-level job satisfaction in the form of spillover externalities.²⁰ The degree of this feedback is larger at the workplace level than local labor market level. Thus, enforcing job satisfaction policies at the workplace level will likely be more effective than implementing such policies at the local labor market level. This result is particularly important, because it is reported in the literature that job satisfaction is positively related to worker productivity [see, for example, Boeckerman and Ilmakunnas (2012)]. In terms of the magnitudes, Boeckerman

²⁰Employers can stimulate social interactions among workers, which suggests that the optimal design of worker groups/teams should also account for these social forces [Tumen (2012)].

and Ilmakunnas (2012) report that one standard deviation increase in group-level job satisfaction raises productivity per hours worked by 6.6 percent. This means that the existence of spillover externalities introduces notable gains to increasing job satisfaction at the individual level.

Estimates for the Contextual Effects. We control for a large set of contextual variables in our regressions. However, only a few of them produce statistically significant coefficients. We start with the WERS results, in which we report estimates for social interactions at the workplace level. Our WERS regressions [see Table (3)] show that the Male, Age, Log Earnings, and Relative Earnings variables are subject to statistically significant contextual social effects. To begin with, we show that working close to a group of workers with a larger fraction of males increases job satisfaction at the workplace-level analysis. This result can be interpreted in several ways. It is well-documented in the literature that females are more likely to be absent from work due to illness-related reasons.²¹ If this is the case, workplace attitudes such as motivation, attachment, and involvement might be weaker for females than males due to these relatively more frequent breaks in their work attendance. As a consequence, working in a group with a greater fraction of males might increase motivation and, thus, job satisfaction. A second explanation might be related to gender discrimination; that is, our finding can be interpreted as the existence of distaste against women. However, we are cautious on this interpretation as we do not have additional empirical support for this claim in our analysis. Apart from the contextual gender effects, we document that job satisfaction is higher in groups with higher average worker age and this positive impact becomes weaker as the average age goes up in our WERS sample. This can be attributed—using the Mincerian language—to labor market experience. Working in a group with a larger fraction of experienced workers may produce external effects boosting job satisfaction and, thus, worker productivity.

We also find that earnings have statistically significant contextual effects in our WERS regressions. The contextual earnings effect refers to the effect of the mean earnings in one's reference group on individual-level job satisfaction. To comply with the conventions in the literature,

²¹Ichino and Moretti (2009) show that this may be related to menstrual cycles.

we construct two earnings variables: (1) the natural logarithm of earnings and (2) a dummy variable indicating the earnings rank of the worker, i.e., relative earnings. As we report in Table (3)], the average earnings in the reference group is negatively related to the individual job satisfaction score. Moreover, working in a group with a greater fraction workers earning more than the median wage also reduces individual job satisfaction.²² This is consistent with the findings in the pay-comparisons literature, which suggest that job satisfaction depends on relative income comparisons [see, for example, Clark et al. (2009) and Card et al. (2012)]. Our findings confirm the view that income is evaluated relative to some comparison level based on the reference group and not only in absolute terms. This is in line with the findings reported in the literature [see, e.g., Easterlin (1973)].

For the BHPS analysis, we do not find any statistically significant contextual effects for the Male, Age, Log Earnings, and Relative Earnings variables at the local labor market level, unlike our workplace level analysis. This may be due to the reason that individuals care less about the group-level exogenous characteristics in larger reference groups. However, we do report a different contextual effect at the local labor market level: the promotion opportunities, a variable that does not exist in the WERS data set. Specifically, we find that working in groups with a greater fraction of workers with access to promotion opportunities reduces individual-level job satisfaction [see Table (4)]. This can be attributed to competition: that is, if there is a large fraction of workers in one's reference group expecting promotions, then this might increase excess competition in the work environment and, therefore, might reduce job satisfaction. It is interesting, however, detecting this result at the local labor market level. This may be suggesting that working in industries with harsher competition conditions reduces individual-level job satisfaction. We do not have results for this variable at the workplace level; however, if the competition hypothesis is true, we conjecture that the contextual effect of promotion opportunities would be even stronger at the workplace level.

²²So, instead of the Hirschman's tunnel effect, we observe that envy/hatred is more likely to be effective.

5.1.2 The Role of Heterogeneity

The discussion above focuses on the marginal effects of the variables and ignores the magnitudes of the interaction terms. But, the interaction terms in the hierarchical model may themselves yield interesting results. In particular, the interaction terms might suggest whether the endogenous and/or contextual social effects exhibit any heterogeneity, i.e., whether the social interactions estimates depend on the environment in which the associated interactions take place. For the sake of brevity, we perform this task only with the WERS data.

Our "social ecologies" model features two types of interaction terms. First, the $m_g \times Y_g$ interaction indicates whether the endogenous social effect differs across groups with different characteristics. For example, the effect of group-level job satisfaction on individual-level job satisfaction may be different in groups with different gender dominance structures or different education compositions. Second, the $X_{i_g} \times Y_g$ interaction indicates whether the individuallevel effects change with respect to group-level characteristics. For example, the relationship between individual-level characteristics and the individual-level job satisfaction scores may be different in groups with different characteristics. Table (5) reports the full set of interaction terms in the first group. The results provide interesting insights on the group-level heterogeneity structure. Most importantly, we find that the endogenous social effects are stronger in firms with a larger fraction of older and more educated workers. This suggests that working in a group of individuals with higher human capital levels exposes workers to stronger peer effects in job satisfaction. The second set of interaction terms include so many variables that it would be inefficient to report all of them. Instead, we prefer to mention verbally the most striking estimates. We find some gender effects that worth mentioning. Table (3) documents that job satisfaction is higher in groups with a greater fraction of male workers. We find that this effect is much larger for male workers than females. We also show that the negative relative income effect is felt much stronger by males than females. These patterns suggest that both endogenous and contextual social effects are subject to a certain degree of heterogeneity in the worker population.

It remains to check whether the interaction terms are jointly statistically significant or not. This is equivalent to test the relevance of our identifying assumption, which introduces non-linearities to the standard linear-in-means model through incorporating interaction terms as in Equation (3.11). If the interaction terms are jointly insignificant, then the model will favor the linear-in-means model over the hierarchical model; therefore, our model will remain fundamentally unidentified and would be subject to Manski's criticism. We test the joint significance of the interaction terms both for the WERS and BHPS data sets using the F-test. For the WERS, the interaction terms are statistically significant at 1 percent level with a p-value very close to zero. For the BHPS, the interaction terms are still jointly statistically significant, but the level of significance is 5 percent (with a p-value around 0.045). This means that the validity of our identifying assumption is roughly confirmed by both of the data sets we use.

5.1.3 Linear-in-Means Model

Although the simple test we perform above favors the hierarchical model over the linear-in-means model and although we are aware of the fact that the linear-in-means models are potentially plagued with the reflection problem, it will still be a useful exercise to get estimates from the linear-in-means model to understand the extent to which the reflection problem could contaminate the estimates. The linear specification is given by Equation (3.1), which says that the individual-level job satisfaction is a linear function of individual-level characteristics, group-level characteristics, and group-level job satisfaction. Again, for the sake of brevity, we perform this exercise only with the WERS data set.

Table (6) reports the linear-in-means estimates. We compare them to the estimates for the hierarchical model given in Table (3). The most striking difference is that the endogenous social effect is 0.423 in the hierarchical model, while it is 0.514 in the linear-in-means model. This difference implies that the linear-in-means model overestimates the endogenous social effect by around 22 percent, which is a non-negligibly large bias. The contextual effects are also different; some differing only in magnitude, some differing in both sign and magnitude.

The punchline is that ignoring the reflection problem has a potential to generate large biases in the estimates of social interactions in job satisfaction. The reason is that when individual-and group-level outcomes are simultaneously determined within groups, then the parameter estimates from the linear-in-means model would just be a combination of the underlying structural parameters and the true social effects will be masked. The hierarchical model offers a way to separately identify the social effects.

5.2 Robustness Checks

In this subsection, we perform several robustness checks for the results that our hierarchical model produces. We carry out this task in two different ways. First, we calculate the group-level means of the variables using an alternative technique and, second, given the flexibility offered by this alternative formulation, we propose a different way of controlling for group-level heterogeneity. It is perhaps needed to mention at this stage that we are severely constrained by data availability in extending the variety of these robustness exercises. Overall, the two alternative exercises presented below strongly suggest that the social interactions estimates obtained from the baseline hierarchical model [see Table (3)] are robust.

5.2.1 An Alternative Formulation of Group-Level Variables

Since the group sizes are, on average, small especially for the WERS data set, one might suspect that the social interactions estimates reported in Table (3) could depend on the way group-level means are calculated. In our baseline calculations, the group-level means are calculated as the simple arithmetic average of the individual-level variables. In particular, our original formulation assumes that the group-level job satisfaction score that affects individual i's job satisfaction level also includes i's own job satisfaction. This has two potential deficiencies. First, if the groups are small, then this formulation will impose a mechanical positive correlation between the individual- and group-level job satisfaction scores, because a change in the individual-level score will be more likely generate a sizable effect on group-level job satisfaction score. Second, it is also plausible to say that the individual i in group g is exposed to a social effect that is characterized by the mean behavior in the group, where the mean is

calculated excluding i. When the groups are large, including or excluding the observation for i in calculating the group-level means should be less of a concern, since the group means will be almost identical in both cases.

Based on these concerns, we reconstruct the group-level averages for each individual, excluding the corresponding individual. We re-estimate the hierarchical model based on these new group-level definitions and the results are presented in Table (7) in the form of marginal effects. The endogenous social effect, which is 0.408, is only slightly smaller than the original estimate of 0.423. The qualitative nature of the contextual effects also remain unchanged, with the exception that the coefficients of the human capital variables are slightly smaller and the coefficients of the absolute and relative earnings variables are somewhat larger in Table (7) relative to the estimates reported in Table (3). These results suggest that our estimates are robust to the changes in the method chosen to calculate the group-level means.

5.2.2 Group Fixed Effects

Another potential problem with our baseline estimates is that group-level unobserved heterogeneity is controlled only by assuming a random-effects structure in the error term. By definition, the random-effects model is based on the assumption that the error term is uncorrelated with other covariates. Although, this specification is useful in controlling for the fact that the error term may exhibit within-group correlations, it would be inadequate in capturing the group-level fixed effects, which are unobserved factors correlated with the determinants of individual- and group-level variables. Fortunately, the alternative calculation procedure introduced in Section 5.2.1 allows us to incorporate the group-level fixed effects manually. Since the group-level variables are calculated excluding i, these variables offer some variation within the group. When such a variation exists, one can generate group dummies and use these dummies as group fixed effects.²³ We perform this task both for the WERS and BHPS

 $^{^{23}}$ To understand this point, suppose that there is a group of 10 workers and each of them has received a score based on a certain test. If you calculate the group-level mean of the test score by including everyone, then the mean test score will be the same for everyone in the group. If the individual i excludes his own test score in calculating the group-level mean he faces, and if everyone does so, then the group-level means will be different for everyone in the group. When there is individual-level variation in the key group-level variables, then it becomes possible to include group-level fixed effects into the regressions. In the absence of such a variation, the group-level fixed effects would be collinear with the group-level means. In particular, the number of omitted firm dummies would be equal to the number of group-level means included into the regression. Thus, the main coefficients of interest would become uninterpretable.

datasets.

For the WERS analysis, we re-estimate the hierarchical model by incorporating these firm fixed effects in a regression in which the group-level variables are calculated excluding the individual *i*. Table (8) demonstrates the results. Strikingly, we find that the endogenous social effect is estimated to be 0.367, which is somewhat lower than but still reasonably close to our original estimate 0.423. This estimate also supports the qualitative nature of our results, which suggests that the social interactions in job satisfaction are strong at the firm level.²⁴ The signs and significance levels of other variables are also mostly remained unaltered—except that the coefficient of the relative income variable turned to insignificant after including group-level fixed effects. At the end, we conclude that our original WERS estimates are robust to the inclusion of firm-level fixed effects.

For the BHPS analysis, we again re-performed the estimations for our hierarchical model after incorporating the industry-region fixed effects, where the group-level variables are calculated excluding the individual i. Table (9) documents the results. The original estimate for the endogenous social effect was 0.147. We find that this estimate declined slightly and became 0.141 after including the group-level fixed effects. Again, the qualitative nature of other coefficients remains unaltered. This confirms that our BHPS estimates are robust too.

5.3 Policy Implications

It will perhaps be useful to put the policy implications of our results in a nutshell. These results suggest that (1) there are large gains to policy interventions to increase individual-level job satisfaction as there are significant positive feedback effects from group-level job satisfaction toward individual-level job satisfaction in the form of spillover externalities; (2) failing to account for the spillover externalities in job satisfaction may lead us to mis-assess the effectiveness of job satisfaction policies; thus, the policy maker should internalize these externalities; and (3) job satisfaction spillovers are much stronger at the workplace level than local labor

²⁴We would like to note that this approach is also not free of any problems. When the group size is small, the accuracy of controlling for group-level heterogeneity via a dummy variable will be low. So, from a "model selection" viewpoint, it is not clear if the model with firm fixed effects should be preferred to the model that excludes the fixed effects.

market level: therefore, designing/enforcing job satisfaction policies at the workplace level will likely be more effective than implementing such policies at the local labor market level.

Although we find that the job satisfaction spillovers are particularly strong at the workplace level, our results also suggest that it would be misleading to exclude the possibility of spillovers in broader reference groups. In other words, regional aspects of working life and conditions in the local labor markets may also induce interactions among people that can exhibit non-negligible social effects.

6 Concluding Remarks

There is a large literature arguing that peer effects exist in various labor market outcomes including productivity, wages, absenteeism, and learning (or knowledge spillovers). We contribute to this literature in three ways. First, this is the first paper in the literature testing the existence of job satisfaction spillovers. We show that there exist significant positive spillovers in job satisfaction. Second, we perform our analysis at two different aggregation levels using two different data sets from the United Kingdom. We find that the job satisfaction spillovers are almost three times stronger at the workplace level than local labor market level (defined in terms of industry \times region cells). This implies that although job satisfaction spillovers are strong among narrowly defined worker groups, it would be misleading to exclude the possibility of spillovers in broader reference groups. In other words, regional aspects of working life and conditions in the local labor markets may also induce interactions among people that can exhibit non-negligible social effects. Finally, we make a methodological contribution to the empirical literature by resolving the identification problem using an intuitive insight from the hierarchical models of social processes. Specifically, we hypothesize that our parameters of interest are determined within the social environments they originate from. Under reasonable specifications, this logic implies introducing certain cross-product terms into the standard estimating equations.

We conclude that there are sizable social interactions in job satisfaction that should not be

ignored in assessing policy effectiveness. The policy makers should internalize these spillover externalities. Our estimates also provide guidance on the question "at which level job satisfaction spillovers should be internalized." We argue that firms should design and implement their own job satisfaction policies rather than relying on more general policies or institutional regulations that could only be enforced at the local labor market level (or industry level).

References

- AKERLOF, G. A., A. K. Rose, and J. Yellen (1988): "Job Switching and Job Satisfaction in the US Labor Market," *Brookings Papers on Economic Activity*, 2, 495–594.
- ANGRIST, J. D. (2014): "Perils of Peer Effects," Forthcoming, Labor Economics.
- Argyle, M. (2001): The Psychology of Happiness, London: Routledge.
- AZOULAY, P., J. S. GRAFF ZIVIN, AND Z. WANG (2010): "Superstar Extinction," Quarterly Journal of Economics, 125, 549–589.
- Bandiera, O., I. Barankay, and I. Rasul (2009): "Social Connections and Incentives in the Workplace: Evidence from Personnel Data," *Econometrica*, 77, 1047–1094.
- Blume, L. E., W. A. Brock, S. N. Durlauf, and Y. M. Ioannides (2011): "Identification of Social Interactions," in *Handbook of Social Economics*, ed. by J. Benhabib, A. Bisin, and M. O. Jackson, New York, NY: Elsevier, 853–964.
- Blume, L. E. and S. N. Durlauf (2001): "The Interactions-Based Approach to Socioe-conomic Behavior," in *Social Dynamics*, ed. by S. N. Durlauf and H. P. Young, Cambridge, MA: MIT Press.
- BOECKERMAN, P. AND P. ILMAKUNNAS (2012): "The Job Satisfaction-Productivity Nexus: A Study Using Matched Survey and Register Data," *Industrial and Labor Relations Review*, 65, 244–262.
- Bramoulle, Y., H. Djebbari, and B. Fortin (2009): "Identification of Peer Effects through Social Networks," *Journal of Econometrics*, 150, 41–55.
- Brock, W. A. and S. N. Durlauf (2001a): "Discrete Choice with Social Interactions," Review of Economic Studies, 68, 235–260.

- Brown, S. P. and R. A. Peterson (1994): "The Effects of Effort on Sales Performance and Job Satisfaction," *Journal of Marketing*, 58, 70–80.
- CARD, D. E., A. MAS, E. MORETTI, AND E. SAEZ (2012): "Inequality at Work: The Effect of Peer Salaries on Job Satisfaction," *American Economic Review*, 102, 2981–3003.
- CAVAN, R. S. (1983): "The Chicago School of Sociology, 1918–1933," *Journal of Contemporary Ethnography*, 11, 407–420.
- Christen, M., G. Iyer, and D. Soberman (2006): "Job Satisfaction, Job Performance, and Effort: A Reexamination Using Agency Theory," *Journal of Marketing*, 70, 137–150.
- CLARK, A. E. (1996): "Job Satisfaction in Britain," *British Journal of Industrial Relations*, 34, 189–217.
- CLARK, A. E., Y. GEORGELLIS, AND P. SANFEY (1998): "Job Satisfaction, Wage Change, and Quits," *Research in Labor Economics*, 17, 95–121.
- CLARK, A. E., N. KRISTENSEN, AND N. WESTERGARD-NIELSEN (2009): "Job Satisfaction and Co-worker Wages: Status or Signal?" *Economic Journal*, 119, 430–447.
- CLARK, A. E. AND A. J. OSWALD (1996): "Satisfaction and Comparison Income," *Journal of Public Economics*, 61, 359–381.
- CORNELISSEN, T., C. DUSTMANN, AND U. SCHOENBERG (2013): "Peer Effects in the Workplace," IZA Discussion Paper No. 7617.
- DEATON, A. (1990): "On Risk, Insurance, and Intra-Village Consumption Smoothing," Unpublished manuscript, Princeton University.

- EASTERLIN, R. A. (1973): "Relative Economic Status and the American Fertility Swing," in *Family Economic Behavior: Problems and Prospects*, ed. by E. B. Sheldon, Philadelphia, PA: Lippincott.
- Falk, A. and A. Ichino (2006): "Clean Evidence on Peer Effects," *Journal of Labor Economics*, 24, 39–58.
- FERRER-I-CARBONELL, A. (2005): "Income and Well-Being: An Empirical Analysis of the Comparison Income Effect," *Journal of Public Economics*, 89, 997–1019.
- FOWLER, J. H. AND N. A. CHRISTAKIS (2008): "Dynamic Spread of Happiness in a Large Social Network: Longitudinal Analysis over 20 Years in the Framingham Heart Study," *British Medical Journal*, 337, a2338.
- Freeman, R. B. (1978): "Job Satisfaction as an Economic Variable," *American Economic Review*, 68, 135–141.
- GHINETTI, P. (2007): "The Public-Private Job Satisfaction Differential in Italy," *Labour*, 21, 361–388.
- GLAESER, E. L., B. I. SACERDOTE, AND J. A. SCHEINKMAN (1996): "Crime and Social Interactions," *Quarterly Journal of Economics*, 111, 507–548.
- ——— (2003): "The Social Multiplier," Journal of the European Economic Association, 1, 345–353.
- Green, F. and N. Tsitsianis (2005): "An Investigation of National Trends in Job Satisfaction in Britain and Germany," *British Journal of Industrial Relations*, 43, 401–430.
- Guryan, J., K. Kroft, and M. J. Notowidigdo (2009): "Peer Effects in the Workplace: Evidence from Random Groupings in Professional Golf Tournaments," *American Economic Journal: Applied Economics*, 1, 34–68.
- Harter, J. K., F. L. Schmidt, and T. L. Hayes (2002): "Business-Unit-Level Relationship between Employee Satisfaction, Employee Engagement, and Business Outcomes: A Meta-Analysis," *Journal of Applied Psychology*, 87, 268–279.

- Hatfield, E., J. Cacioppo, and R. L. Rapson (1994): *Emotional Contagion*, New York, NY: Cambridge University Press.
- IAFFALDANO, M. T. AND P. M. MUCHINSKY (1985): "Job Satisfaction and Job Performance: A Meta-Analysis," *Psychological Bulletin*, 97, 251–273.
- ICHINO, A. AND G. MAGGI (2000): "Work Environment and Individual Background: Explaining Regional Shirking Differentials in a Large Italian Firm," Quarterly Journal of Economics, 115, 1057–1090.
- ICHINO, A. AND E. MORETTI (2009): "Biological Gender Differences, Absenteeism, and the Earnings Gap," American Economic Journal: Applied Economics, 1, 183–218.
- JACKSON, C. K. AND E. BRUEGMANN (2009): "Teaching Students and Teaching Each Other: The Importance of Peer Learning for Teachers," American Economic Journal: Applied Economics, 1, 85–108.
- Jones, M. K., R. Jones, P. L. Latreille, and P. J. Sloane (2009): "Training, Job Satisfaction, and Workplace Performance in Britain: Evidence from WERS 2004," *Labour*, 23, 139–175.
- JONES, M. K. AND P. J. SLOANE (2010): "Disability and Skill Mismatch," *Economic Record*, 86, 101–114.
- JUDGE, T. A., C. J. THORESEN, J. E. BONO, AND G. K. PATTON (2001): "The Job Satisfaction-Job Performance Relationship: A Qualitative and Quantitative Review," Psychological Bulletin, 127, 376–407.
- KNIGHT, J. AND R. GUNATILAKA (2009): "Is Happiness Infectious?" Unpublished manuscript, Oxford University.
- LUTTMER, E. F. P. (2005): "Neighbors as Negatives: Relative Earnings and Well-Being," Quarterly Journal of Economics, 120, 963–1002.
- Manski, C. F. (1993): "Identification of Endogenous Social Effects: The Reflection Problem," *Review of Economic Studies*, 60, 531–542.

- MAS, A. AND E. MORETTI (2009): "Peers at Work," American Economic Review, 99, 112–145.
- MOFFITT, R. (2001): "Policy Interventions, Low-Level Equilibria, and Social Interactions," in *Social Dynamics*, ed. by S. N. Durlauf and P. Young, Cambridge, MA: MIT Press.
- Mumford, K. and P. N. Smith (2009): "What Determines the Part-time and Gender Earnings Gaps in Britain: Evidence from the Workplace," Oxford Economic Papers, 61, 56–75.
- ———— (2013): "Gender, Job Satisfaction, and Relative Wages," Unpublished manuscript, University of York.
- OSTROFF, C. (1992): "The Relationship between Satisfaction, Attitudes, and Performance: An Organizational Level Analysis," *Journal of Applied Psychology*, 77, 963–974.
- OSWALD, A. J., E. Proto, and D. Sgroi (2013): "Happiness and Productivity," *Journal of Labor Economics*, forthcoming.
- Otis, N. and L. G. Pelletier (2005): "A Motivational Model of Daily Hassles, Physical Symptoms, and Future Work Intentions among Police Officers," *Journal of Applied Social Psychology*, 35, 2193–2214.
- Patterson, M., P. Warr, and M. West (2004): "Organizational Climate and Company Productivity: The Role of Employee Affect and Employee Level." *Journal of Occupational and Organizational Psychology*, 77, 193–216.
- RAUDENBUSH, S. W. AND R. J. SAMPSON (1999): "Econometrics: Toward a Science of Assessing Ecological Settings, with Application to the Systematic Social Observation of Neighborhoods," *Sociological Methodology*, 29, 1–41.
- Ryan, A. M., M. J. Schmit, and R. Johnson (1996): "Attitudes and Effectiveness: Examining Relations at an Organizational Level," *Personnel Psychology*, 49, 853–882.

- Sato, W. and S. Yoshikawa (2007): "Spontaneous Facial Mimicry in Response to Dynamic Facial Expressions," *Cognition*, 104, 1–18.
- Schneider, B., P. J. Hanges, D. B. Smith, and A. N. Salvaggio (2003): "Which Comes First: Employee Attitudes or Organizational Financial and Market Performance?" Journal of Applied Psychology, 88, 836–851.
- SLOANE, P. J. AND H. WILLIAMS (2000): "Job Satisfaction, Comparison Earnings, and Gender," *Labour*, 14, 473–502.
- SOETEVENT, A. R. (2006): "Empirics of the Identification of Social Interactions: An Evaluation of the Approaches and Their Results," *Journal of Economic Surveys*, 20, 193–228.
- STUTZER, A. AND R. LALIVE (2004): "The Role of Social Work Norms in Job Searching and Subjective Well-Being," *Journal of the European Economic Association*, 2, 696–719.
- TAYLOR, M. P. (2006): "Tell Me Why I Don't Like Mondays: Investigating Day of the Week Effects on Job Satisfaction and Psychological Well-Being," *Journal of the Royal Statistical Society (Series A)*, 169, 127–142.
- Tumen, S. (2012): "A Theory of Intra-firm Group Design," Unpublished manuscript, University of Chicago.
- Tumen, S. and T. Zeydanli (2014): "Is Happiness Contagious? Separating Spillover Externalities from the Group-Level Social Context," Forthcoming, *Journal of Happiness Studies*.
- Waldinger, F. (2012): "Peer Effects in Science: Evidence from the Dismissal of Scientists in Nazi Germany," *Review of Economic Studies*, 79, 838–861.
- Wegge, J., K. Schmidt, C. Parkes, and K. van Dick (2007): "Taking a Sickie': Job Satisfaction and Job Involvement as Interactive Predictors of Absenteeism in a Public Organization," *Journal of Occupational and Organizational Psychology*, 80, 77–89.
- Zelenski, J. M., S. A. Murphy, and D. A. Jenkins (2008): "The Happy-Productive Worker Thesis Revisited," *Journal of Happiness Studies*, 9, 521–537.

Variable	Mean	$\mathbf{Std}.\mathbf{Dev}$	
Job satisfaction	4.20	2.11	
Std. job satisfaction	1.07e-08	1	
Individual-level Characteristics			
Male	0.469	0.499	
Married	0.682	0.466	
Age	41.646	12.095	
Higher degree	0.009	0.092	
First degree	0.033	0.18	
'A'-level	0.273	0.446	
'O'-level	0.238	0.426	
Vocational qual.	0.446	0.497	
Log earnings	5.692	0.74	
Relative earnings	0.518	0.5	
Working hours	0.656	0.475	
Job-level Characteristics			
Private sector	0.365	0.481	
Public sector	0.551	0.497	
Union membership	0.462	0.499	
Small-size establishment	0.324	0.468	
Medium-size establishment	0.318	0.466	
Large-size establishment	0.358	0.48	
# of observations	20,035		
# of workplaces/establishments	1,673		

Table 1: Summary Statistics – WERS. Workplace Employment Relations Survey 2004 data set is used to construct this table.

Variable	Mean	Std.Dev
Job satisfaction	5.40	1.26
Std. job satisfaction	-3.13e-09	1
Individual-level Characteristics		
Male	0.474	0.499
Married	0.573	0.495
Age	40.351	12.049
Higher degree	0.044	0.205
First degree	0.158	0.365
'A'-level	0.126	0.332
'O'-level	0.176	0.381
Other higher qual.	0.308	0.462
Vocational qual.	0.094	0.292
No qual.	0.094	0.291
Log earnings	6.907	0.707
Relative earnings	0.531	0.499
Job-level Characteristics		
Union membership	0.331	0.412
Promotion opportunities	0.499	0.5
Full-time job	0.79	0.407
Private sector	0.63	0.48
Public sector	0.22	0.42
Prefer to work fewer hours	0.328	0.47
Prefer to work more hours	0.054	0.226
Prefer to contain same hours	0.618	0.486
Small-size establishment	0.486	0.5
Medium-size establishment	0.216	0.411
Large-size establishment	0.299	0.458
Industries	0.200	0.100
Energy-water supplies	0.036	0.186
Extraction-manufacture	0.074	0.262
Metal goods-engineering	0.046	0.21
Other manufacturing	0.058	0.235
Construction	0.178	0.383
Distribution, hotels, catering	0.096	0.295
Transport-communication	0.196	0.397
Banking-finance	0.254	0.435
Other services	0.061	0.239
Regions	0.001	0.200
London	0.05	0.219
South East	0.114	0.318
South West	0.054	0.226
East Anglia	0.024	0.154
East Midlands	0.0521	0.222
West Midlands	0.046	0.209
Northwest	0.066	0.249
Yorkshire-Humberside	0.054	0.245 0.226
North East	0.034 0.037	0.220
Wales	0.037 0.148	0.169 0.355
Scotland	0.148 0.188	0.391
Northern Ireland	0.166	0.391 0.372
# of observations		128
# Of Observations	0,4	±40

Table 2: Summary Statistics – BHPS. British Household Panel Survey 2004 cross-section is used to construct this table.

MARGINAL EFFECTS – Hierarchical model				
	Individual level		Group level	
Covariate	Coefficient	(St. Error)	Coefficient	(St. Error)
Endogenous social effect				
Mean job satisfaction (standardized)	_	_	0.423***	(0.007)
Individual characteristics				
Male	-0.160***	(0.017)	0.157***	(0.038)
Married	0.063***	(0.016)	-0.075	(0.051)
Age	-0.031***	(0.005)	0.026**	(0.013)
Age-squared/100	0.039***	(0.005)	-0.034**	(0.015)
First degree	0.147	(0.161)	0.094	(0.341)
'A'-level	0.265*	(0.150)	-0.073	(0.307)
'O'-level	0.298**	(0.500)	-0.099	(0.305)
Vocational qual.	0.430***	(0.149)	-0.211	(0.302)
Log earnings	0.232***	(0.022)	-0.203***	(0.032)
Relative earnings	0.083***	(0.021)	-0.093*	(0.057)
Working hours	-0.076***	(0.020)	0.047	(0.044)
Establishment/job characteristics	3			
Private sector	_	_	-0.003	(0.028)
Public sector	_	_	-0.004	(0.027)
Not union member	_	_	-0.015	(0.030)
Medium-size establishment	_	_	-0.003	(0.017)
Large establishment	_	_	-0.006	(0.017)
# of observations	20,035			

Table 3: ESTIMATION RESULTS (WERS). *, **, *** indicate the 10%, 5%, and 1% significance levels, respectively. Standard errors, clustered at the group (e.g., firm) level, are reported in parentheses. Group-level unobserved effects are controlled for. The first two columns document the estimates for individual-level coefficients, while the last two columns document those for group-level coefficients (e.g., the social interactions estimates). The coefficients should be interpreted as the "marginal effects." Since the individual- and group-level variables are the same for WERS (i.e., the characteristics of the firm that the worker is employed is also the firm- or group-level characteristics) we drop the individual-level coefficients and report only group-level estimates for the establishment/job characteristics.

MARGINAL EFFECTS – Hierarchical model Individual level Group level Covariate Coefficient (St. Error) Coefficient (St. Error) Endogenous social effect 0.147***Mean job satisfaction (standardized) (0.028)Individual characteristics Male -0.172*** (0.030)0.328(0.244)0.161*** Married (0.027)0.368(0.303)-0.028*** Age (0.007)0.094(0.070)Age-squared/100 0.039*** (0.009)-0.144*(0.084)Higher degree 0.262**(0.102)-1.033(0.771)0.194**First degree (0.099)-1.048(0.742)'A'-level 0.142(0.091)-0.701(0.602)'O'-level 0.235**(0.094)-0.889(0.671)Other higher qual. 0.137(0.096)-0.888(0.788)Vocational qual. 0.052(0.094)-1.152(0.790)0.119*** Log earnings (0.035)-0.198(0.193)Relative earnings 0.111*** (0.038)-0.233(0.423)Job characteristics Private sector -0.025(0.083)0.196(0.272)Public sector -0.127(0.096)0.082(0.287)Union membership -0.060*(0.033)0.178(0.200)0.282*** Promotion opportunities (0.026)-0.430* (0.236)-0.183*** Full-time job (0.044)-0.244(0.445)-0.393*** Prefer to work fewer hours (0.027)0.276(0.279)-0.181*** Medium-size establishment (0.033)0.310 (0.337)-0.162*** Large establishment (0.033)-0.052(0.273)6,428 # of observations

Table 4: ESTIMATION RESULTS (BHPS). *, **, *** indicate the 10%, 5%, and 1% significance levels, respectively. Standard errors, clustered at the group level, are reported in parentheses. Group-level unobserved effects are controlled for. The first two columns document the estimates for individual-level coefficients, while the last two columns document those for group-level coefficients (e.g., the social interactions estimates). The coefficients should be interpreted as the "marginal effects."

INTERACTION TERMS $(m_g \times Y_g)$ – Hierarchical model				
Covariate	Coefficient	(St. Error)		
Individual characteristics				
$m_g \times Y_g \text{ (Male)}$	0.012	(0.085)		
$m_g \times Y_g \text{ (Married)}$	0.011	(0.103)		
$m_g \times Y_g \text{ (Age)}$	0.044*	(0.024)		
$m_g \times Y_g \text{ (Age-squared/100)}$	-0.049*	(0.029)		
$m_g \times Y_g$ (First degree)	0.928*	(0.467)		
$m_g \times Y_g$ ('A'-level)	0.847*	(0.416)		
$m_g \times Y_g$ ('O'-level)	0.864	(0.567)		
$m_g \times Y_g$ (Vocational qual.)	0.710*	(0.386)		
$m_g \times Y_g$ (Log earnings)	0.039	(0.054)		
$m_g \times Y_g$ (Relative earnings)	-0.119	(0.096)		
$m_g \times Y_g$ (Working hours)	-0.151*	(0.083)		
Establishment/job characteristics				
$m_g \times Y_g$ (Private sector)	0.116	(0.081)		
$m_g \times Y_g$ (Public sector)	0.042	(0.078)		
$m_g \times Y_g$ (Not union member)	0.127*	(0.065)		
$m_g \times Y_g$ (Medium-size establishment)	-0.122**	(0.049)		
$m_g \times Y_g$ (Large establishment)	-0.079	(0.053)		
# of observations	20,	035		

Table 5: Group-Level interaction terms (WERS). *, **, *** indicate the 10%, 5%, and 1% significance levels, respectively. Standard errors, clustered at the group (e.g., firm) level, are reported in parentheses. Group-level unobserved effects are controlled for.

Linear-in-means model	·	·		
	Individual level		Group level	
Covariate	Coefficient	(St. Error)	Coefficient	(St. Error)
Endogenous social effect				
Mean job satisfaction (standardized)	_	_	0.514***	(0.020)
Individual characteristics				
Male	-0.141***	(0.017)	-0.031	(0.030)
Married	0.083***	(0.016)	0.076**	(0.037)
Age	-0.029***	(0.004)	0.008	(0.009)
Age-squared/100	0.039***	(0.005)	-0.009	(0.011)
First degree	-0.062	(0.080)	-0.231	(0.212)
'A'-level	0.031	(0.072)	-0.277	(0.178)
'O'-level	0.039	(0.074)	-0.445**	(0.176)
Vocational qual.	0.170**	(0.074)	-0.439**	(0.173)
Log earnings	0.136***	(0.020)	-0.099***	(0.026)
Relative earnings	0.137***	(0.021)	-0.016	(0.046)
Working hours	-0.066***	(0.020)	0.031	(0.034)
Establishment/job characteristics	8			
Private sector	_	_	-0.018	(0.020)
Public sector	_	_	-0.044**	(0.018)
Not union member	_	_	-0.102***	(0.021)
Medium-size establishment	_	_	-0.032**	(0.013)
Large establishment	_	_	-0.070***	(0.014)
# of observations	20,035			

Table 6: ESTIMATION RESULTS (WERS): Linear-in-means model - *, **, *** indicate the 10%, 5%, and 1% significance levels, respectively. Standard errors, clustered at the group (e.g., firm) level, are reported in parentheses. Group-level unobserved effects are controlled for. The first two columns document the estimates for individual-level coefficients, while the last two columns document those for group-level coefficients (e.g., the social interactions estimates). Since the individual- and group-level variables are the same for WERS (i.e., the characteristics of the firm that the worker is employed is also the firm- or group-level characteristics) we drop the individual-level coefficients and report only group-level estimates for the establishment/job characteristics.

MARGINAL EFFECTS – Hierarchical model with group means excluding i .				
	Individ	ual level	Group level	
Covariate	Coefficient	(St. Error)	Coefficient	(St. Error)
Endogenous social effect				
Mean job satisfaction (standardized)	_	_	0.408***	(0.008)
Individual characteristics				
Male	-0.143***	(0.021)	0.181***	(0.041)
Married	0.086***	(0.020)	-0.069	(0.053)
Age	-0.038***	(0.006)	0.028**	(0.014)
Age-squared/100	0.044***	(0.006)	-0.036**	(0.016)
First degree	0.122	(0.172)	0.083	(0.366)
'A'-level	0.192	(0.177)	-0.084	(0.357)
'O'-level	0.243**	(0.487)	-0.109	(0.352)
Vocational qual.	0.380**	(0.195)	-0.254	(0.342)
Log earnings	0.245***	(0.021)	-0.211***	(0.033)
Relative earnings	0.097***	(0.022)	-0.096*	(0.058)
Working hours	-0.066***	(0.018)	0.039	(0.046)
Establishment/job characteristics	5			
Private sector	_	_	-0.001	(0.032)
Public sector	_	_	-0.005	(0.029)
Not union member	_	_	-0.028	(0.034)
Medium-size establishment	_	_	0.004	(0.022)
Large establishment	_	_	-0.002	(0.021)
# of observations	20,035			

Table 7: ESTIMATION RESULTS (WERS): Group means exclude i-*, **, *** indicate the 10%, 5%, and 1% significance levels, respectively. Standard errors, clustered at the group (e.g., firm) level, are reported in parentheses. Group-level unobserved effects are controlled for. The first two columns document the estimates for individual-level coefficients, while the last two columns document those for group-level coefficients (e.g., the social interactions estimates). The coefficients should be interpreted as the "marginal effects." Since the individual- and group-level variables are the same for WERS (i.e., the characteristics of the firm that the worker is employed is also the firm- or group-level characteristics) we drop the individual-level coefficients and report only group-level estimates for the establishment/job characteristics.

Large establishment

Firm fixed effects

of observations

MARGINAL EFFECTS – Hierarchical model with group means excluding i.

Firm fixed effects are included. Individual level Group level Covariate Coefficient (St. Error) Coefficient (St. Error) Endogenous social effect 0.367*** Mean job satisfaction (standardized) (0.026)Individual characteristics -0.096*** 0.149*** Male (0.026)(0.051)0.081*** Married (0.025)-0.043(0.062)-0.032*** Age (0.010)0.023*(0.015)0.035*** (0.011)-0.027* Age-squared/100 (0.017)First degree 0.099 (0.192)0.052(0.392)'A'-level 0.161(0.197)-0.061(0.422)'O'-level 0.203 (0.381)-0.089(0.386)Vocational qual. 0.320*(0.191)-0.199(0.401)0.211*** -0.188*** Log earnings (0.027)(0.057)0.091*** (0.024)-0.085(0.072)Relative earnings -0.056** Working hours (0.029)0.029(0.051)Establishment/job characteristics Private sector (0.037)-0.001Public sector 0.012(0.046)Not union member -0.041(0.066)Medium-size establishment -0.016(0.052)

Table 8: ESTIMATION RESULTS (WERS): Group means exclude i, firm fixed effects included - *, ***, **** indicate the 10%, 5%, and 1% significance levels, respectively. Standard errors, clustered at the group (e.g., firm) level, are reported in parentheses. Group-level unobserved effects are controlled for. The first two columns document the estimates for individual-level coefficients, while the last two columns document those for group-level coefficients (e.g., the social interactions estimates). The coefficients should be interpreted as the "marginal effects." Since the individual- and group-level variables are the same for WERS (i.e., the characteristics of the firm that the worker is employed is also the firm- or group-level characteristics) we drop the individual-level coefficients and report only group-level estimates for the establishment/job characteristics. Firm-level fixed effects are included.

0.007

Included

20.035

(0.053)

${\bf MARGINAL\ EFFECTS-Hierarchical\ model}$

Industry-region cell fixed effects are included.

	Individual level		Group level	
Covariate	Coefficient	(St. Error)	Coefficient	(St. Error)
Endogenous social effect				
Mean job satisfaction (standardized)	_	_	0.141***	(0.031)
Individual characteristics				
Male	-0.151***	(0.036)	0.216	(0.301)
Married	0.146***	(0.031)	0.221	(0.441)
Age	-0.027***	(0.008)	0.054	(0.091)
Age-squared/100	0.037***	(0.010)	-0.066	(0.097)
Higher degree	0.251**	(0.107)	-1.201	(0.894)
First degree	0.183*	(0.111)	-1.103	(0.801)
'A'-level	0.110	(0.103)	-0.755	(0.792)
'O'-level	0.216*	(0.124)	-0.904	(0.687)
Other higher qual.	0.109	(0.122)	-1.001	(0.904)
Vocational qual.	0.021	(0.127)	-1.202	(0.905)
Log earnings	0.111***	(0.037)	-0.123	(0.240)
Relative earnings	0.104***	(0.040)	-0.175	(0.501)
Job characteristics				
Private sector	-0.029	(0.084)	0.101	(0.303)
Public sector	-0.110	(0.106)	0.044	(0.401)
Union membership	-0.029	(0.041)	0.099	(0.386)
Promotion opportunities	0.211***	(0.052)	-0.211	(0.388)
Full-time job	-0.141***	(0.054)	-0.178	(0.489)
Prefer to work fewer hours	-0.288***	(0.051)	0.200	(0.476)
Medium-size establishment	-0.149***	(0.037)	0.234	(0.356)
Large establishment	-0.141***	(0.036)	-0.012	(0.298)
Industry-region fixed effects	Included			
# of observations	6,428			

Table 9: ESTIMATION RESULTS (BHPS): Group means exclude i, industry-region cell fixed effects included – *, **, *** indicate the 10%, 5%, and 1% significance levels, respectively. Standard errors, clustered at the group (e.g., industry-region) level, are reported in parentheses. Group-level unobserved effects are controlled for. The first two columns document the estimates for individual-level coefficients, while the last two columns document those for group-level coefficients (e.g., the social interactions estimates). The coefficients should be interpreted as the "marginal effects." Group-level fixed effects are included.